Author's response to reviews

Title: Comorbid mental disorders in substance users from a single catchment area - a clinical study.

Authors:

Anne-Marit Langås (anne-marit.langas@vestreviken.no)
Ulrik F Malt (u.f.malt@medisin.uio.no)
Stein Opjordsmoen (s.e.o.ilner@medisin.uio.no)

Version: 4 Date: 6 February 2011

Author's response to reviews: see over
To the Editor
BMC Psychiatry

Kongsberg 06.02.2011

MS: 1358476400447128 – Comorbid mental disorders in substance users from a single catchment area – a clinical study.

Dear Editor,

Thank you for your acceptance of our manuscript.

The following formatting changes have been made:
- The list of authors is written exactly as in the submission system.
- All highlighting in the manuscript is removed.
- The section ‘Supervision, collaboration, and networks’ is renamed ‘Acknowledgement’.
- The manuscript has been checked for typographical errors and final corrections have been made.

Kind regards,

Anne-Marit Langås
MD / PhD research fellow